
diff -r c45d3cd9a749 -r dcaab8068baa handouts/pep-ho.tex
--- a/handouts/pep-ho.tex	Sat Nov 04 16:17:19 2017 +0000
+++ b/handouts/pep-ho.tex	Sun Nov 05 12:02:09 2017 +0000
@@ -8,7 +8,7 @@
 
 \begin{document}
 
-\section*{A Crash-Course on Scala}
+\section*{A Crash-Course in Scala}
 
 \subsection*{The Very Basics}
 
@@ -40,7 +40,7 @@
 
 \noindent indicating that the result of the addition is of type
 \code{Int} and the actual result is 5; \code{res0} is a name that
-Scala gives automatically to the result. Yoy can reuse this name later
+Scala gives automatically to the result. You can reuse this name later
 on. Another classic example you can try out is
 
 \begin{lstlisting}[numbers=none]




